Author’s response to reviews

Title: Sports participation related to injuries and illnesses among ambulatory youth with chronic diseases: results of the Health in Adapted Youth Sports study

Authors:
Kristel Lankhorst (kristel.lankhorst@hu.nl)
Janke de Groot (janke.degroot@hu.nl)
Tim Takken (T.Takken@umcutrecht.nl)
Frank Backx (F.J.G.Backx@umcutrecht.nl)

Version: 2 Date: 18 Jun 2019

Author’s response to reviews:

Thank you for the clear comments based on our submission. We apologize for any inconvenience, regarding the delay in submitting and reviewing. Herewith you find our response to the editor comments.

Editor Comments:

1. In the Ethics approval and consent to participate subsection of the Declarations please clarify whether consent from the participants and/or their legal proxies was obtained in written or verbal form. Response: We have clarified the informed consent of the participants and their legal proxies in the ethics approval section as follows; Consent from the participants and/or their legal proxies was obtained in written form. See page 16.

2. Please note that the Consent for publication statement refers to consent obtained to include any individual person’s data in any form (including individual details, images or videos). Therefore, please state "Not applicable" in this section as no personal information is provided in your manuscript. Response: We thank the editor for this comment. We have now stated; Not applicable, at the consent for publication section. See page 16.

The statement regarding authors’ approval of the final manuscript should instead be included in the Authors’ contributions section of the Declarations.

Response: We thank the editor for this comment. We have rephrased the Authors' contributions section as follows: All authors have the appropriate permissions and rights to the reported data in this manuscript. All authors gave their consent of this manuscript to be published in Journal of Sports Science, Medicine
Rehabilitation. All authors have agreed to authorship and order of authorship for this manuscript. See page 17.